**Author’s response to reviews**

**Title:** Randomised controlled trial to improve health and reduce substance use in established psychosis (IMPaCT): Cost-effectiveness of integrated psychosocial health promotion

**Authors:**

Margaret Heslin (margaret.heslin@kcl.ac.uk)

Anita Patel (anita.patel@qmul.ac.uk)

Daniel Stahl (daniel.r.stahl@kcl.ac.uk)

Poonam Gardner-Sood (poonam.sood@kcl.ac.uk)

Manyara Mushore (mushorem@lsbu.ac.uk)

Shubulade Smith (shubulade.smith@kcl.ac.uk)

Kathy Greenwood (k.e.greenwood@sussex.ac.uk)

Oluwadamilola Onagbesan (Oluwadamilola.Onagbesan@kcl.ac.uk)

Conan O'Brien (conan.obrien@kcl.ac.uk)

Catherine Fung (Catherine.Fung@elft.nhs.uk)

Ruth Ohlsen (ruth.ohlsen@kcl.ac.uk)

David Hopkins (dhopkins3@nhs.net)

Philippa Lowe (lowephilippa@aol.com)

Maurice Arbutnot (mauricea@easy.com)

Stanley Mutsatsa (Stanley.mutsatsa@city.ac.uk)

Gill Todd (gillian.todd27@btinternet.com)

Anna Kolliakou (anna.Kolliakou@kcl.ac.uk)

John Lally (john.lally@kcl.ac.uk)

Brendon Stubbs (brendon.stubbs@kcl.ac.uk)

Khalida Ismail (khalida.2.ismail@kcl.ac.uk)
Anthony David (anthony.david@kcl.ac.uk)
Robin Murray (robin.murray@kcl.ac.uk)
Zerrin Atakan (zerrin.atakan@kcl.ac.uk)
Fiona Gaughran (fiona.p.gaughran@kcl.ac.uk)

Version: 2 Date: 29 Aug 2017

Author’s response to reviews:

The clinical paper has been accepted to your journal now so if this linked economic paper was excepted it would be really useful for them to be published in the same issue.

1. This has been modified to: The datasets generated and/or analysed during the current study are not publicly available due to them containing information that could compromise research participant privacy/consent but suitably anonymised sub-sections can be made available from the corresponding author on reasonable request.

2. Participants provided written informed consent. This detail has been added to the 'Ethics and consent to participate' declaration.

3. The figure legend has been moved from the figure file to the manuscript. The tables have also been moved to the manuscript.

4. The authors contribution has been rewritten to:

Authors’ Contributions

AP, DS, SS, KG, DH, GT, KI, AD, RM, ZA and FG conceived the study. MH conducted all data analysis with oversight from AP. MH and AP drafted the original manuscript. All authors contributed to the critical review and revision of the manuscript. All authors read and approved the final manuscript.

5. Clean manuscript attached.